Author's response to reviews

Title: Perforation of Meckel's Diverticulum caused by a chicken bone: a case report

Authors:

Kim W Chan - (kim.chan@nhs.net)

Version: 3  Date: 20 October 2008

Author's response to reviews:

This is an unusual presentation of perforated Meckel's Diverticulum caused by chicken bone. Besides, it's a useful topic for discussion among junior surgeon in training as diagnosis of perforated Meckel's diverticulum can be easily missed as acute appendicitis. The correct diagnosis will lead to the correct management. Although only 2% of the population has Meckel's diverticulum, there is still a significant number of patient presented with lower abdominal pain caused by complications from Meckel's diverticulum such as haemorrhage, obstruction or perforation.

Another point to stress for this case report is the unusual perforation caused by chicken bone. This is a good case report to be published for other junior surgeon in training to have a better idea of the potential complications of Meckel's Diverticulum. As such, I would like you to consider seriously the educational value of this case report.

the manuscript was seen and was asked to revised for its language and I have done the revision and also updated some evidence for the manuscript. I hope that the revised manuscript will be better and be forwarded for peer review. I have also omitted part of the sentences as commented by one of the reviewer. In addition, I have also revised the languages in the paper thoroughly. As such, I hope that this case report can be forwarded for further review.